Author’s response to reviews

Title: Validation of the Kidney Disease Quality of Life-Short Form: a cross-sectional study of a dialysis-targeted health measure in Singapore

Authors:

Veena D Joshi (joshiveena0@gmail.com)
Nandakumar Mooppil (nandakumar.m@nkfs.org)
Jeremy FY Lim (jeremy.lim.f.y@sgh.com.sg)

Version: 5 Date: 10 December 2010

Author’s response to reviews:

Dear Editor,

The document (the latest version of revisions) have been formatted as follows.

Major revisions

Authors’ contributions: We have used initials instead of full name.

It is indicated that all authors have read and approved the final manuscript.

Tables: Tables have been numbered in ascending numerical order from 1 to 5 and any references to them in the text have been changed accordingly.

References: References have been cited in the text using consecutive numbers in square brackets. All the references are cited in consecutive numerical order in the manuscript.

Minor revisions

Affiliations: We have included the country in the affiliation details.

Highlighting/tracking: We have removed all highlighting and tracking from the manuscript.

Typography: We have checked the manuscript for any typographical errors.

Thank you,

Veena Joshi